Author's response to reviews

Title: The Edinburgh Type 2 Diabetes Study: Study Protocol

Authors:

Jackie F Price (Jackie.Price@ed.ac.uk)
Rebecca M Reynolds (rreynold@staffmail.ed.ac.uk)
Rory J Mitchell (R.J.Mitchell@ed.ac.uk)
Rachel M Williamson (Rachel.Williamson@luht.scot.nhs.uk)
Gerry F R Fowkes (Gerry.Fowkes@ed.ac.uk)
Ian J Deary (I.Deary@ed.ac.uk)
Amanda J Lee (A.J.Lee@abdn.ac.uk)
Brian M Frier (Brian.Frier@ed.ac.uk)
Peter C Hayes (P.Hayes@ed.ac.uk)
Mark W J Strachan (Mark.Strachan@luht.scot.nhs.uk)

Version: 2 Date: 27 November 2008

Author's response to reviews:

Details of the 'Authors' contributions' have been added as requested to the revised manuscript.